Author's response to reviews

Title: Healthy Hearts - a community-based primary prevention programme to reduce coronary heart disease

Authors:

Gill Richardson (Gill.Richardson@nphs.wales.nhs.uk)
Hugo van Woerden (hvw@doctors.org.uk)
Lucy Morgan (Lucy.Morgan@nphs.wales.nhs.uk)
Rhiannon Edwards (Rhiannon.Edwards@nphs.wales.nhs.uk)
Monica Harries (Monica.Harries@nphs.wales.nhs.uk)
Elaine Hancock (ehancock@sky.com)
Susan Sroczynska (Susan.Sroczynska@nphs.wales.nhs.uk)
Mererid Bowley (mererid.bowley@nphs.wales.nhs.uk)

Version: 5 Date: 14 June 2008

Author's response to reviews: see over
Dear Editor

My apologies for the delay in getting back to you. The changes you have requested have been made.

In particular, the title page has been added including full names and affiliations.

The background section of the abstract has been extended to place the study in its context. The abstract on the submission page has also been updated.

A declaration has been made indicating that the authors have no competing interests.

The figures have also been cropped.

Please let me know if you wish any other changes to be made.

Yours sincerely

Hugo van Woerden